

Water Resources Research[®]

RESEARCH ARTICLE

10.1029/2020WR029194

Key Points:

- Canopy density effect on peak snow accumulation and snowpack duration varies with winter climates
- Canopy thinning is most effective in wet/warm winter climates for improving snowpack duration
- Greatest decrease in snowpack duration under warming is anticipated for snowpack under dense canopy in presently warm winter climates

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

N. Sun, ning.sun@pnnl.gov

Citation:

Sun, N., Yan, H., Wigmosta, M. S., Lundquist, J., Dickerson-Lange, S., & Zhou, T. (2022). Forest canopy density effects on snowpack across the climate gradients of the western United States mountain ranges. *Water Resources Research*, 58, e2020WR029194. https:// doi.org/10.1029/2020WR029194

Received 6 NOV 2020 Accepted 23 DEC 2021

Author Contributions:

Conceptualization: Ning Sun, Hongxiang Yan, Mark S. Wigmosta, Jessica Lundquist, Susan Dickerson-Lange, Tian Zhou Data curation: Ning Sun Formal analysis: Ning Sun, Hongxiang Yan Funding acquisition: Mark S. Wigmosta Investigation: Ning Sun Methodology: Ning Sun, Hongxiang Yan, Mark S. Wigmosta, Jessica Lundquist, Susan Dickerson-Lange, Tian Zhou Project Administration: Mark S. Wigmosta Supervision: Mark S. Wigmosta Validation: Ning Sun Visualization: Ning Sun, Hongxiang Yan Writing - original draft: Ning Sun Writing - review & editing: Ning Sun, Hongxiang Yan, Mark S. Wigmosta, Jessica Lundquist, Susan Dickerson-Lange, Tian Zhou

© 2022. American Geophysical Union. All Rights Reserved.

Forest Canopy Density Effects on Snowpack Across the Climate Gradients of the Western United States Mountain Ranges

Ning Sun¹, Hongxiang Yan¹, Mark S. Wigmosta^{1,2}, Jessica Lundquist², Susan Dickerson-Lange³, and Tian Zhou⁴

¹Energy and Environment Directorate, Pacific Northwest National Laboratory, Richland, WA, USA, ²Department of Civil and Environmental Engineering, University of Washington, Seattle, WA, USA, ³Natural Systems Design, Seattle, WA, USA, ⁴Earth and Biological Sciences Directorate, Pacific Northwest National Laboratory, Richland, WA, USA

Abstract Controlled field experiments to disentangle the effect of canopy density from the effect of climate on snowpack dynamics are limited by the underlying linkage between canopy density and climate. Thus, based on observations alone, it is not well understood how variations in canopy density can affect snow processes under different climate regimes. To address this knowledge gap, this study uses a physics-based modeling approach to evaluate the sensitivity of snowpack dynamics to variations in canopy density across the climate gradients of the Western U.S. as represented by 228 Snow Telemetry (SNOTEL) sites. Within the model, we uniformly parameterize the canopy across sites to represent an idealized forest with high, medium, and low canopy density, respectively. The results illustrate that the effect of canopy density on the peak snow water equivalent (SWE) and duration of under-canopy snowpack is sensitive to winter climate (i.e., climatological winter precipitation and temperature). As canopy density decreases, the greatest increase in peak SWE and snowpack duration is found in wet/warm and dry/cold climates, where snowpack under low-density forest lasts longer than that in the open. In comparison, peak SWE and snowpack duration in wet/cold climates are less sensitive to changing canopy density. Thus, forest management actions (e.g., thinning and clearing) are likely to have disparate impacts on snow depending on local winter climate. Climate sensitivity of under-canopy snowpack suggests that snowpack duration under dense canopy in presently warm winter climates is expected to experience the greatest reduction under a warming climate.

Plain Language Summary In the Western U.S., forest management (e.g., clearing and thinning) has great potential for altering snow processes, which is particularly important for locations susceptible to low flows in the summer. With limited observations available to evaluate climate-forest-snow relationships, there is a lack of knowledge about how changing forest density can affect snow water storage and snowpack duration under different climate conditions. To address this gap, this study uses a physics-based snow model to simulate snow processes under changing canopy density for over 200 locations representative of the Western U.S. climate diversity. Results suggest that the effect of canopy density on snowpack water storage and duration varies with winter climate, and thus the effect of forest management on snow and water resources can vary substantially with climate. In wet/warm and dry/cold climates, a decrease in canopy density generally increases snowpack duration such that a low-density forest keeps snowpack longer than an open area in the same location. From the forest management perspective, canopy thinning in these climates provides better chances than forest clearing for enhancing snowpack duration and storage. In wet/cold climates, snowpack duration is less sensitive to changing canopy density.

1. Introduction

Snow cover dynamics can have strong implications for water supply (Bales et al., 2006; Barnett et al., 2005; Mote et al., 2005; Sturm et al., 2017), flood risk (Hamlet & Lettenmaier, 2007; Sharma et al., 2018; Yan, Sun, Wigmosta, Skaggs, Hou, & Leung, 2019; Yan, Sun, Wigmosta, Skaggs, Hou, Leung, et al., 2019), ecosystem function (Coughlan & Running, 1997; Wheeler et al., 2016), and economies (Sturm et al., 2017). In mountain forests, snowpack dynamics can be markedly impacted by canopy cover characteristics through canopy snow interception and sublimation, and influencing snowpack energy balance (Cristea et al., 2014; Essery et al., 2008, 2009; Hawthorne et al., 2013; Lundquist et al., 2013; Pomeroy et al., 2009; Storck et al., 2002). Given the opportunities of managing forests for snowpack retention and sustaining low flows, a growing number of field studies (e.g., Dickerson-Lange et al., 2017; Harpold et al., 2015; Lawler & Link, 2011; Mazzotti, Currier, et al., 2019; Murray & Buttle, 2003; Musselman et al., 2008; Roth & Nolin, 2017; Veatch et al., 2009; Winkler et al., 2005) have been conducted to examine how forest impacts the snow regime in comparison to its neighboring open site. These field observations provide valuable insights and are essential for parameterizing and evaluating models related to canopy snow processes. For example, many field studies confirm that under-canopy peak snow water equivalent (SWE) is lower than that in the nearby open area, which is attributed largely to canopy interception that could account for up to 83% of snowfall (Martin et al., 2013). Field observations collected in maritime climates generally indicate that snow lasts longer in the open than the forest, except for locations influenced by wind-driven snow deposition or redistribution (Dickerson-Lange et al., 2017; Lundquist et al., 2013; Roth & Nolin, 2017; Rutter et al., 2009; Storck, 2000). On the other hand, there are disagreements in field observations conducted in other climates. For example, sensors collecting multi-year snow depth data at Boulder Creek, Colorado suggested longer snow duration in small forest openings compared to under-canopy areas (Harpold et al., 2015), while other studies (Rutter et al., 2009; Thyer et al., 2004) observed longer snow duration under the forest in similar cold and dry winter climates.

A key challenge in field studies is creating controlled experiments that vary only the climate while holding all forest characteristics fixed for determining the inter-related sensitivity of climate versus canopy on snow processes. In nature, canopy characteristics often covary with climate due to biomass-climate correlations on large scales such that temperate and boreal forests with greater biomass (i.e., taller trees and denser growth) occur generally in warmer and wetter climates (Keith et al., 2009; Liu et al., 2014; Stegen et al., 2011). Thus, the conclusions from aggregated observational studies (e.g., Lundquist et al., 2013) that snow lasts longer in the open in areas with warmer winters could be due to either the climate or to the denser forests associated with that climate. Moreover, field and satellite observations are generally constrained by local weather and canopy characteristics, which makes it inherently difficult to rely on field data alone to disentangle interactions between climate, snow and forest across climate gradients. Most snow observations also do not have long enough records to cover the range of interannual climate variability, which can interact with canopy to enhance or diminish the canopy effect on snow processes. For example, snow measurements collected over 17 snow seasons in northwestern Russia showed shifting effects of forest canopy from increasing to decreasing snow duration compared to the paired open area (Gelfan et al., 2004). Thus, field studies alone, are not fully able to disentangle the effect of canopy density from the effect of climate on snow processes.

To address the knowledge gap, we applied a process-based snow model of the Distributed Hydrology-Soil-Vegetation Model (DHSVM, Wigmosta et al., 1994) to examine, based on model physics, how we expect varying canopy density to impact snow processes and the inter-related sensitivity of snow processes to climate and canopy density. Model experiments were conducted for 228 Snow Telemetry (SNOTEL) sites to represent the climate gradients of the Western U.S. mountain ranges. For each SNOTEL site, we configured DHSVM for an idealized forest with a range of canopy density, including high, medium and low canopy density, and a reference open condition (with no canopy cover and no canopy influence on snow processes). Canopy density is defined in the model in a two-dimensional sense, based on the canopy fractional coverage (FC) and the leaf area index (LAI). For each canopy density category, uniform canopy parameters were assumed across sites to allow for comparisons of snowpack sensitivity to changing climates, without confounding them with the heterogeneity in in situ canopy characteristics.

2. Data and Methods

2.1. DHSVM Snow Model Physics

DHSVM is a physics-based hydrological model that was initially developed by Wigmosta et al. (1994) for modeling overland and subsurface hydrological processes governed by the mass and energy balance at each model pixel of 10–150 m at a subdaily timescale. DHSVM simulates accumulation and melt of ground snowpack using a two-layer mass and energy balance model within each model pixel that is treated as a single, uniform snowpack. Ground snowpack is represented by a thin surface layer and a deep pack layer. Energy exchange between the atmosphere, overstory canopy, and snowpack is simulated by the energy balance at the snow surface:

$$Q = \text{NSW} + \text{NLW} + H + \text{LE} + M \tag{1}$$





Figure 1. Distributed Hydrology-Soil-Vegetation Model (DHSVM) representation of a model pixel with a fractional coverage of forest canopy (FC). Forest canopy is characterized by leaf area index (LAI) and canopy height (h).

where Q is net energy input to snowpack, NSW is net shortwave radiation, NLW is net longwave radiation, H is sensible heat flux, LE is latent heat flux, and M is advected heat from rainfall to snowpack. Ground heat flux and heat conduction between the surface and pack layer are neglected given that they generally represent minor heat contribution to melt when compared to the other snowpack energy balance terms (Marks & Dozier, 1992). Mass and energy exchange between the surface layer and the pack layer occurs via the exchange of ice and meltwater. Excessive liquid water in the pack layer above its liquid water holding capacity is released to the underlying soil as snowpack outflow.

In the presence of overstory canopy, DHSVM partitions a model pixel into a canopy-covered fraction (prescribed by FC) and an open fraction (i.e., 1-FC, Figure 1). The model assumes no interaction between the open and forest fraction such as shading and wind attenuation from forest canopy. Each snowpack energy balance term (Equation 1) is calculated separately for the forest and open fraction, which is then weighted by FC to calculate the pixel-level snowpack energy balance. Forest canopy is characterized primarily by LAI and canopy height. LAI that varies monthly is used to represent spatially aggregated small openings in forest canopy, and sub-pixel variability

of forest elements such as spatial patterns of small openings is neglected. Generally, lower LAI values result in less canopy snow interception, more under-canopy shortwave radiation, and less under-canopy down-welling longwave radiation.

Snow-canopy processes are simulated by a one-layer mass and energy balance canopy model. At the pixel level, *NSW* received by snowpack is the area-weighted average of attenuated shortwave radiation from the canopy fraction and direct shortwave radiation in the open fraction:

$$NSW = (1 - \alpha)(R_s(1 - FC) + FC(R_{dir}\tau_b + R_{diff}\tau_d))$$
(2)

where α is the snowpack albedo, which is estimated as a function of snow surface temperature and the days since last snowfall. Details of snow albedo curve parameterization and calibration are provided by Sun et al. (2019). R_s is incoming shortwave radiation and is partitioned into direct radiation (R_{dir}) and diffuse radiation (R_{diff}) based on the clearness index (Sun et al., 2015; Wigmosta et al., 1994), and τ_b is the fraction of direct shortwave radiation transmitted through canopy, which is estimated by the Beer-Bouger-Lambert law (Peixoto & Oort, 1992):

$$\tau_b = e^{-k \cdot \frac{h}{\sin\theta}} \tag{3}$$

where *h* is canopy height, θ denotes solar elevation angle, and *k* is monthly radiation extinction coefficient estimated as a linear function of LAI (Sun et al., 2015). NLW received by the snowpack is estimated by:

$$NLW = L_d(1 - FC) + FC \cdot \sigma T_c^4 - \sigma T_s^4$$
(4)

where L_d is downward longwave radiation emitted from the atmosphere, σ is the Stephan-Boltzmann constant, T_c is the canopy temperature approximated by air temperature in Kelvin, and T_s is snow surface temperature in Kelvin. Snowfall can be intercepted by canopy up to the maximum interception capacity, which is estimated as a function of LAI and air temperature. Intercepted snow can be removed from canopy through snowmelt, sublimation, and mass release. Melt of intercepted snow is simulated by the energy balance approach. When snowmelt exceeds the water holding capacity of canopy snowpack, meltwater drips from canopy and is added to the ground snowpack as rain. Mass release occurs if sufficient snow is available and is estimated as a linear function of meltwater drip. Rain, snowfall not intercepted by canopy is combined with drip and mass release contribute mass and energy to the ground snowpack. The readers are referred to Wigmosta et al. (1994, 2002), Storck et al. (1998), and Andreadis et al. (2009) for more details of DHSVM snow model physics.





Figure 2. Climate gradients of the Western U.S. mountain ranges, represented by 228 SNOTEL locations, which are classified into four winter climate zones (wet/warm, dry/warm, wet/cold, and dry/cold) based on their winter (DJF) temperature and precipitation.

2.2. Study Domain and BCQC SNOTEL Data

Previously, Sun et al. (2019) calibrated and validated the DHSVM snow model to the daily SWE measurements at 246 SNOTEL stations assuming an entirely open condition and flat terrain. Their site selection was based largely on the quality and duration of meteorological and SWE measurements. Specifically, Sun et al. (2019) screened the raw daily measurements of temperature, precipitation and SWE from 805 active SNOTEL stations, using the rigorous three-stage quality control to identify missing data, remove erroneous values and outliers as described in detail by Yan et al. (2018). The screened data were then bias corrected for warm air temperature bias and snowfall undercatch following the approach described by Sun et al. (2019). The resulting data set is referred to as the Bias-Corrected Quality-Controlled (BCQC) SNOTEL data set, which includes daily records of precipitation, maximum and minimum air temperature, and SWE for 805 stations with varying durations.

From the BCQC SNOTEL data set, we found 246 SNOTEL stations with continuous meteorological and snow records over the longest common period, which was 2007–2013. We then chose 228 SNOTEL stations with a Nash-Sutcliffe Efficiency (NSE, Nash & Sutcliffe, 1970) of simulated daily SWE > 0.65. Over these 228 stations, the simulated NSE is >0.8 at 98.7% of the locations, and the overall mean NSE is 0.94. The percentage error in simulated peak SWE is within $\pm 15\%$ at 88.3% of the stations, and the over-

all mean error is -7.0%; the bias in simulated snow disappearance date (SDD) is within 7 days at 77.9% of the stations, and the overall mean bias is 4.1 days. Here SDD is defined as the day when the modeled SWE is below 5 mm (Serreze et al., 1999), and the moving average of SWE over its subsequent 14 days is below 5 mm. The latter criterion avoids erroneous identifications of SDD when intermittent melt-out occurs due to limited snowfall and/or warm winters. For details of model configuration, parameterization and calibration procedures, the reader is referred to Sun et al. (2019).

We divided the 228 SNOTEL stations into four climate classes based on their climatological averages of winter (December through February, or DJF) temperature and precipitation, wet/warm, dry/warm, wet/cold, and dry/ cold (Figure 2). By definition, winter temperature $<-1^{\circ}$ C is categorized as cold winter climate, and winter precipitation <300 mm (~ 25 th percentile of winter precipitation distribution across sites) as dry winter climate. For example, sites with winter temperature $<-1^{\circ}$ C and winter precipitation >300 mm are categorized as wet/ cold. With this classification, only one SNOTEL station was classified as dry/warm. It should be noted that the categorization into simplified climate classes provides a framework to consider groupings of results, but results are presented across the spectrum of temperature and precipitation values.

2.3. Model Configuration and Experiments

Using the same snow parameters calibrated by Sun et al. (2019), we configured the snow model at each SNOTEL site to represent a range of changing canopy density, the forest with high, medium, and low canopy density, and the reference open condition. The canopy density is distinguished by two-dimensional parameter values including canopy fractional coverage and LAI as described in Table 1. As there are not consistent thresholds in the literature defining canopy density classes, we chose three variations of parameters that bracket the range of the canopy parameters reported in the literature to represent changing canopy densities (Dickerson-Lange et al., 2017;

Table 1

Distributed Hydrology-Soil-Vegetation Model (DHSVM) Parameters Used to Characterize the Open and Forest With High, Medium, and Low Canopy Density

| Model parameters | Open | High-density forest | Medium-density forest | Low-density forest | Description | |
|---|------|---------------------|-----------------------|--------------------|---|--|
| Fractional Coverage | 0% | 95% | 70% | 40% | Fraction of ground area covered by forest overstory | |
| LAI (unit: m ² /m ²) | - | 8 | 4.5 | 2 | Monthly canopy leaf area index | |

Gelfan et al., 2004; Harpold et al., 2015; Lundquist et al., 2013; Martin et al., 2013; North et al., 2004; Roth & Nolin, 2017; Rutter et al., 2009; Storck, 2000; Thomas & Winner, 2000; Thyer et al., 2004). Because the effect of canopy height is secondary to canopy density (see Figures S1–S5 in Supporting Information S1), we use a 40 m canopy height in following analyses that is representative of mature conifer in the Pacific Northwest.

For each SNOTEL site, the model was run at a 3-hourly time step forced by meteorological inputs including precipitation, air temperature, relative humidity, downward shortwave radiation, downward longwave radiation, and wind speed. All meteorological variables but wind speed were obtained from disaggregating daily meteorological records (i.e., maximum and minimum daily air temperature, and precipitation) from the BCQC SNOTEL data set using the MTCLIM algorithms (Bohn et al., 2013; Maurer et al., 2002; Thornton & Running, 1999). Subdaily precipitation presumed daily precipitation occurred at a uniform rate throughout the day. Subdaily air temperatures were estimated based on daily minimum and maximum air temperatures using a spline of third-order Hermite polynomials. Relative humidity was calculated from subdaily temperatures assuming that the daily minimum temperature is equal to the dew point. Downward shortwave radiation was calculated based on daily temperature range and dewpoint temperature using the Thornton and Running (1999) algorithm, where dewpoint temperature was estimated based on daily minimum temperature and precipitation. Downward longwave radiation was calculated as a function of subdaily temperatures using the method described by Prata (1996). Wind speed, which was not collected at SNOTEL stations, was taken from the wind speed field of a gridded meteorological data set (Livneh et al., 2013). As noted previously, the modeling intention is not to realistically represent the in situ forest cover or forest gap conditions, which are highly heterogeneous across sites. The goal is to examine, based on modeled physics, how peak SWE and snowpack duration change under varied forest density across the climate gradients. For selected sites representative of climate gradients, we also analyzed snowpack energy balance components to identify the key processes that determine the snowpack dynamics as canopy density changes.

2.4. Climate Sensitivity Analysis

Luce et al. (2014) revealed strong correlation between the climatology of winter temperature and precipitation and the climatology of snowpack storage across the Western U.S. Of particular interest to this study is how the climate-snow relationship varies as canopy density changes. Following the approach descried in Luce et al. (2014), we used their local polynomial regression approach to examine how climatological averages of winter temperature and precipitation influence the climatology of peak SWE and SDD across sites, as forest cover condition changes. In this case, interannual variations of climate and snowpack, as a result of climate variability at sites were not analyzed. For different canopy density categories, we also analyzed the sensitivities of average peak SWE and SDD to changing winter temperature and precipitation, respectively. For example, the temperature sensitivity of peak SWE measures how average peak SWE changes as a function of the temperature values of all sites. The local polynomial regression and the sensitivity analysis were conducted using the locfit package in R (Loader, 2020). Details about the approach can be found in Luce et al. (2014).

3. Results

3.1. Climate Sensitivity of Under-Canopy Snowpack

For open to any canopy density (ranging from no trees to high-density forest), local polynomial regression analysis (Figure 3) suggests a strong interaction between winter precipitation and temperature in their effects on peak SWE (with NSE values ranging from 0.87 to 0.90). As canopy density increases, peak SWE becomes slightly less sensitive to changing winter temperature and precipitation (Figure 4). The average peak SWE sensitivity to temperature over all sites for the open, low-, medium- and high-density forest is -45.4, -44.5, -43.8, and -39.4 mm/°C, respectively. The average peak SWE sensitivity to precipitation for the open, low-, medium- and high-density forest is 1.6, 1.5, 1.4, and 1.3 mm/mm, respectively. Despite the canopy density, temperature sensitivity of peak SWE is closely related to the precipitation conditions. For example, as demonstrated in Figure 4a, peak SWE at wetter locations shows greater sensitivity to changing temperature. Similarly, precipitation sensitivity of peak SWE also varies with the temperature condition. Generally, colder locations show greater precipitation sensitivity (Figure 4b). Among all locations, peak SWE in the wet/warm climates shows the greatest sensitivity





Figure 3. Peak snow water equivalent (SWE) as a function of December through February (DJF) precipitation and temperature. Contours are fitted surface for the data at SNOTEL sites using local polynomial regression. Units of contour levels are mm.



Figure 4. Sensitivity of peak snow water equivalent (SWE) to changing December through February (DJF) temperature (T; unit: mm-SWE/°C) and changing DJF precipitation (P; unit: mm-SWE/mm-P) calculated as partial derivatives of the contours shown in Figure 3 for (a–b) the open, (c–d) the low-density forest, (e–f) the medium-density forest, and (g–h) the high-density forest.





Figure 5. Snow disappearance date (SDD) as a function of December through February (DJF) precipitation and temperature. Values of contour levels indicate the number of days since the start of a water year (i.e., Oct. 1).

to temperature change, and the greatest sensitivity of peak SWE to precipitation change is found in wet/cold climates.

As canopy density increases, the interaction between winter precipitation and temperature generally decreases in its effect on SDD, indicated by decreasing NSE values from 0.79 to 0.60 (Figure 5). For example, SDD sensitivity to temperature under the high-density forest (Figure 6g) is quite similar across all precipitation ranges, indicated by contour lines more parallel to the precipitation axis. In comparison, for the open condition (Figure 6a), contour lines are more sloped across the range of precipitation and temperature, suggesting a joint sensitivity of SDD to both precipitation and temperature. Similar for the SDD sensitivity to precipitation, as canopy density increases, it becomes less related to the temperature conditions, suggested by more vertical contour lines to the temperature axis, as shown in Figure 6h relative to Figure 6b. In contrast, under the open and low-density forest conditions, the greatest SDD sensitivity over all sites is generally greater as canopy density increases. Average SDD sensitivity to temperature for the open, low-, medium- and high-density forest is -6.5, -6.3, -7.9, and -9.2 days/°C, respectively. SDD sensitivity to precipitation for the open, low-, medium- and high-density forest is 0.08, 0.08, 0.12, and 0.16 days/mm, respectively.

3.2. Canopy Density Effect on Peak SWE

For each canopy density class, we calculated the relative difference in the mean annual peak SWE between the reference open condition and under the canopy (Figure 7), quantified by $\Delta \text{peakSWE}$ (Equation 5). Given the modeling uncertainties, the relative difference was considered insignificant if $\Delta \text{peakSWE}$ is within ±10%. As only one SNOTEL station was classified as dry/warm, the following analyses do not include dry/warm climates.

$$\Delta \text{peakSWE} = \frac{\text{peakSWE(open)} - \text{peakSWE(forest)}}{\text{peakSWE(open)}}$$
(5)



Figure 6. Sensitivity of snow disappearance date (SDD) to changing December through February (DJF) temperature (T; unit: day/°C) and changing DJF precipitation (P; unit: day/mm) calculated as partial derivatives of the contours shown in Figure 5 for (a–b) the open, (c–d) the low-density forest, (e–f) the medium-density forest, and (g–h) the high-density forest.

As expected, peak SWE decreases as canopy density increases across all sites. Compared to the open, the peak SWE difference is most pronounced in the high-density forest. Simulations suggest that peak SWE in the open is always higher than that in the high-density forest by 13%–78% (Figure 7a). For 75 (or 33%) SNOTEL sites located in wet/warm and dry/cold winter climates, peak SWE in the open is higher by more than 50% than in the high-density forest. Compared to the medium-density forest, peak SWE is higher in the open by 10%–28% at about 50% of the locations and is similar in magnitude (within $\pm 10\%$) at the remaining locations (Figure 7b). In contrast, peak SWE in the low-density forest is similar to that in the open at over 80% of the locations, and interestingly, can be higher than the open at some locations, mostly in dry/cold climates (Figure 7c).

To identify the primary process controlling Δ peakSWE under different climates, as canopy density changes, we decomposed simulated daily peak SWE into: (a) cumulative snow gain (positive changes in daily SWE), and (b) cumulative snow loss (negative changes in daily SWE). Comparing the open and the forest, the difference in their cumulative snow gain until the time of peak SWE is denoted by Δ accumulation that is dominated by differences caused by canopy interception, and the difference in cumulative snow loss is denoted by Δ ablation that is dominated by differences in melt rates. As illustrated in Figure 8a, the differential peak SWE between the high-density forest and the open is dominated by Δ accumulation under all climate conditions. As forest density decreases, the magnitude of Δ accumulation decreases primarily due to reduced canopy interception, while Δ ablation at more locations turns from negative values (i.e., higher melt rate in the forest) into positive values (i.e., higher melt rate in the open, Figures 8b and 8c). For about 57% of the locations, Δ accumulation under the lower-density canopy reduces the differential peak SWE.

3.3. Canopy Density Effect on SDD

Simulated mean SDD over 2007–2013 was compared between the open and the forest of varying densities. As in Lundquist et al. (2013), the differential SDD (Δ SDD, Equation 6) within ±3 days is considered negligible.

$$\Delta SDD = SDD(open) - SDD(forest)$$

(6)





Figure 7. Relative difference in peak snow water equivalent (SWE) (Δ peakSWE defined in Equation 5) between the open and (a) high-density forest, (b) medium-density forest, and (c) low-density forest as a function of December through February (DJF) precipitation and DJF temperature. Δ peakSWE within $\pm 10\%$ is considered negligible and is marked by the " \times " symbol. Each plot area is divided into four climate zones: zone I is dry/warm, zone II is dry/cold, zone III is wet/warm, and zone IV is wet/cold.

Overall, the effect of canopy density on Δ SDD varies with winter climates (Figure 9). Snowpack lasts longer in the open than in the high-density forest $(\Delta SDD > 0)$ at 58% of the locations by up to 55 days, mostly characterized by the wet/warm and dry/cold winter climates. Conversely, snowpack lasts longer in the high-density forest (Δ SDD < 0) at 42% of the locations by up to 31 days, mostly in the wet/cold climate. As forest density decreases, more locations (predominantly in the wet/warm and dry/cold climates) show earlier snow disappearance in the open relative to the forest (Δ SDD < 0); the magnitude of Δ SDD generally decreases, suggesting smaller differences in snowpack duration between the open and forest. Comparing snowpack in the medium-density forest to the open, snowpack stays longer in the forest at 60% of the locations by up to 25 days, and lasts longer in the open at only 16% of the locations, mostly in wet/warm or very dry/cold locations, by up to 19 days (Figure 9b). Comparing SDD in the low-density forest to the open, about 90% of the locations show early snow disappearance in the open by up to 27 days (Figure 9c).

To identify the processes controlling Δ SDD, we compared Δ accumulation and Δ ablation (defined in Section 3.2) at the time of first snow disappearance in either the open or under the forest (referred to as the first SDD hereafter). Comparing the high-density forest to the open, analysis (Figure 10a) suggests that Δ SDD in the wet/warm climate is dominated by Δ accumulation. This is consistent with the snow observations collected at several wet/warm locations in the Pacific Northwest by Dickerson-Lange et al. (2017). In the wet/cold (Figure 10d) and dry/cold (Figure 10g) climates, the controlling mechanism of Δ SDD varies from site to site, showing no strong correlation with winter meteorology. As forest density decreases, generally leading to smaller Δ accumulation, Δ SDD decreases for most locations. As canopy density decreases, the controlling mechanism of Δ SDD at more locations shifted from Δ accumulation to Δ ablation (e.g., Figures 10h and 10i), irrespective of winter climate regimes. Analysis of Δ SDD between the low-density forest and the open indicates that Δ melt is the dominant mechanism at all locations.

3.4. Canopy Density Effect on Snowpack Energy Balance

To understand the canopy density effect on some key components of the snowpack energy balance, we examined the monthly snowpack energy balance for three locations representative of the wet/warm, dry/cold, and wet/ cold climate, respectively (Table 2). Two consecutive water years were analyzed including a wetter year 2011 and a drier year 2012. Year 2011 is also slightly colder than 2012 based on mean air temperature over the energy balance analysis period from November to May.

SWE simulations for the wet/warm site (Figure 11a) suggest that, regardless of canopy density and interannual climate variability, there is a significant amount of early melt (prior to peak SWE) that is over 35% of total snow-fall. Compared to the open condition, the amount of early melt under varied canopy density is similar in year 2011 but is notably lower in the drier year 2012. This is because the melt energy is consistently greater in the open than that under the forest of varying densities during the snow accumulation season of year 2012 (Figure 12a). As a result, the difference in peak SWE between the open and forest is relatively smaller in year 2012 because slightly greater early melt in the open offsets the difference in accumulation due to canopy interception (Figures 11a). Unlike the wet/warm site, the dry/cold and wet/cold sites show negligible early melt (Figures 11b and 11c) for both open and forested conditions, due to nearly zero or negative net energy available for melt during DJF (Table 3). Hence, the differential peak SWE between the open and the forest in the dry/cold and wet/cold climates is dominated by the amount of canopy interception, which decreases as canopy density decreases. For all three





Figure 8. Difference in snow accumulation (Δ accumulation) versus difference in snow ablation (Δ ablation) at the time of peak snow water equivalent (SWE) between the open and (a) high-density forest, (b) medium-density forest, and (c) low-density forest. The SNOTEL locations are grouped by their winter climate condition indicated by different colors of circles.

sites, snowmelt after peak SWE is consistently higher in the open than under the forest due to higher melt energy available in the open during the melt season (Table 3). With lower canopy density, melt rates generally increase. This is because, when solar elevations are high, shortwave radiation dominates the snowpack melt energy (Figure 12, Table 3).

4. Discussion

In this study, we provide a model-based assessment of canopy density effects on peak SWE and snow disappearance dates across the climate gradients of the Western U.S. mountain ranges. By design, rather than trying to represent in situ forest cover conditions, uniform, generic canopy characteristics were used across sites for snowpack simulations under canopy density scenarios. Doing so allows for cross-site comparisons of climate sensitivity of snowpack, without confounding them with heterogeneity in local canopy conditions.

Snow measurements from SNOTEL stations were used to evaluate site-level model-simulated SWE for the open condition. While this open condition assumption may not be appropriate for SNOTEL stations located in forest gaps, we consider the natural variability of the SNOTEL siting to be part of the uncertainty in the snow model calibration, which contributes to the spread in the parameter values across all sites. Although it may be an interesting analysis to carefully examine model parameters at each site as a function of its location relative to adjacent topography and vegetation, such an analysis is beyond the scope of this study. For snowpack simulations using hypothetical canopy characteristics, we compare our model results to published snow data, in particular, paired open-forest field observations, to assess general model performance in terms of the open versus forest difference in peak SWE and SDD. In Table S1 in Supporting Information S1, we provide a synopsis of recent studies across diverse climate regimes that provide paired open-forest snow observations as well as general information about forest canopy such as LAI and FC. In general, our findings are consistent with the reported field values. For example, longer snow duration in the open than high-density forest in wet/warm climates (Dickerson-Lange et al., 2017; Storck, 2000), and significantly higher peak SWE in the open than the forest in wet/warm climate (e.g., Andreadis et al., 2009; Dickerson-Lange et al., 2017; Lundquist et al., 2013; Martin et al., 2013; Roth & Nolin, 2019; Westrick et al., 2002). For dry/cold climates, our simulations suggest mixed signs of Δ SDD consistent with observed values in Table S1 in Supporting Information S1. Nevertheless, assumptions used in this analysis and limitations inherent to DHSVM modeling of ground snowpack (discussed in Sun et al., 2019) and canopy snow processes warrant some discussion.

Three-dimensional characteristics of canopy structure such as stem density, stem arrangement, or vertical variation in leaf area are not represented in DHSVM. Canopy density is represented in the model by two-dimensional parameters, FC and LAI. Modifications of these parameters affect snow accumulation and ablation, as well as the total snowpack at a given time step, which is calculated as the average of the under-canopy snowpack





Figure 9. Differential snow disappearance date (SDD) (Δ SDD, unit: day) between the open and the (a) high-density forest, (b) medium-density forest, and (c) low-density forest as a function of December through February (DJF) precipitation and DJF temperature. A darker brown color indicates a longer duration in the open; Δ SDD within ±3 days is marked by the "×" symbol. Zone I is dry/warm, zone II is dry/cold, zone III is wet/warm, and zone IV is wet/cold.

and the open snowpack, as weighted by the fractional forest coverage. By varying these two parameters, the modeling framework simulates the effect of a range of canopy density values at an aggregated spatial scale (i.e., proportions of open and forest) for a gradient of climate conditions. While we test only three variations in canopy density, they are able to bracket the thresholds at which forest presence switches from accelerating to delaying snow disappearance compared to the open across a range of winter temperature and precipitation values. For instance, as forest density increases from zero (i.e., the open condition) to the low density class, the duration of snowpack becomes longer under all climates (Δ SDD < 0). As canopy density increases, snowpack duration becomes more sensitive to climate regimes (e.g., opposite signs of Δ SDD for wet/cold vs. wet/warm climates as demonstrated in Figure 9).

We acknowledge that spatial arrangements of sub-pixel scale forest elements, such as edges between forested and open areas and the size and frequency of canopy gaps, are important for capturing sub-pixel snow distribution and variability. For example, Mazzotti, Malle, et al. (2019) used high resolution Light Detection and Ranging (lidar) observations of snow depth and forest metrics to demonstrate a continuous relation between snow depth and distance from a canopy edge rather than a binary under-canopy versus open classification. Mazzotti et al. (2020, 2021) showed the importance of properly representing the spatial arrangement of canopies and also representing snow covered fraction within relatively small grid cells (50 m). For instance, Mazzotti et al. (2021) showed that using bulk canopy metrics that do not account for the subgrid canopy structure could result in biased estimates of peak SWE. However, it is yet difficult to characterize the fine-scale canopy structure broadly, in great part due to the lack of such data across climatic gradients. Because forest-snow field observations spanning climates and forest types are not cheap, the modeling results presented here provide a clear organizing framework for prioritizing locations and types of future field observations and identifying locations that could benefit from fine-scale canopy snow measurements and modeling.

Wind-related snow preferential deposition, wind-induced unloading of intercepted snow and subsequent redistribution are not represented by the model, which are important to represent especially for simulating snow accumulation at locations with high wind speeds and locations in dry/cold climates with lower snow cohesion (Friesen et al., 2015; Kobayashi, 1987). We recommend improving model representation of wind processes and looking

at specific forest arrangement as a function of climate as a subject for future work. This research notes the need for long-term paired open-forest snow measurements with better control for climate and canopy conditions, particularly for locations with greater sensitivity of snowpack to variations in climate and/or canopy density, for example, the dry/cold and wet/warm sites. Additional observations of sub-canopy longwave and shortwave radiation, such as Malle et al. (2019) and Mazzotti, Malle, et al. (2019), and turbulent fluxes could help constrain and validate parameters for snowpack energy balance. Lastly, uncertainties exist in this analysis associated with model parameterization of canopy snow processes. Given the close interrelation between processes (e.g., canopy loading, unloading, and sublimation) controlling under-canopy snowpack, interactions between parameters need to be carefully examined in uncertainty analysis. Despite limitations in DHSVM modeling and simulations, it should be noted that sensitivity experiments with physics-based models offer an avenue to isolate climate impacts from canopy density impacts on snow processes, which is not possible based on field observations alone.





Figure 10. Differences in snow accumulation (Δ accumulation) versus ablation (Δ ablation) at the time of first snow disappearance date (SDD) between the open and forest of varied density at the (a–c) wet/warm, (d–f) wet/cold, and (g–i) dry/ cold climate. The SNOTEL locations are grouped by Δ SDD. Green circles indicate a longer duration in the forest, and brown circles indicate a longer duration in the open.

5. Conclusions

Due to the covarying nature of climate and canopy density, field observations alone are not yet able to fully isolate the impact of canopy density from the impact of climate on snowpack dynamics. To address this gap, this study uses a physically consistent modeling approach to examine the effects of canopy density variations, due to location, disturbance, or forest management activities, on peak snow accumulation and snow disappearance timing across the climate gradients of the Western U.S.

Local polynomial regression analysis reveals strong interactions between winter precipitation and winter temperature in their effects on peak SWE for the open and forest with varying canopy density. As canopy density increases, the interacting effect of winter precipitation and temperature on SDD generally decreases. Despite

| Table 2 Selected Sites for Snowpack Mass and Energy Balance Analyses | | | | | | |
|--|----------|-----------|------------------|--|--|--|
| Climate | Latitude | Longitude | Location | | | |
| Wet/warm | 42.08 | -123.34 | Bigelow Camp, OR | | | |
| Dry/cold | 44.98 | -111.95 | Short Creek, MT | | | |
| Wet/cold | 39.76 | -107.36 | Bison Lake, CO | | | |

the density of forest canopy, deep snowpack with late snow disappearance is found mostly in wet/cold climates, while shallow snowpack with early snow disappearance appears mostly in warm or dry climates. Climate sensitivity analysis of under-canopy snowpack suggests that canopy density has little effect on the sensitivity of peak SWE to changing winter climate (including both precipitation and temperature). SDD, on the other hand, shows an increasing climate sensitivity as canopy density increases. Despite the canopy density, peak SWE in the wet/warm sites is likely to experience the largest



Water Resources Research



Figure 11. For three sites representative of (a) wet/warm, (b) dry/cold, and (c) wet/cold climate: time series of snow water equivalent (SWE), which are decomposed into Δ accumulation and \sum ablation for water years 2011–2012. \sum ablation are plotted in negative values to be better distinguished from Δ accumulation. Vertical dashed blue lines in the lower subplots indicate the timing of peak SWE.

decrease as climate warms, and wet/cold sites likely experience the largest decrease in peak SWE as precipitation decreases. For the high- and medium-density forest, SDD temperature sensitivity is greatest in warm climates (despite the dry/wet condition), and its precipitation sensitivity is greatest in dry climates (despite the warm/cold condition); under the open and lower-density forest, greatest SDD sensitivity to both temperature and precipitation is expected in dry/warm climates.

Results suggest that the canopy density effect on peak SWE and SDD varies with winter climate. Relative to wet/cold climates, reducing canopy density in wet/warm and dry/cold winter climates shows a greater effect on peak SWE and SDD. In particular, as canopy density decreases in wet/warm and dry/cold climates, snowpack shows a switch from lasting longer in the open to lasting longer in the forest. This is because, as canopy density decreases, the energy-driven snowmelt becomes the dominant influence on snowpack duration, relative to canopy interception during snow accumulation. The modeling framework and results presented here offer an organizing framework that can be used to support forest management that considers how managing forest can improve the resiliency of snow and water resources at different locations. For example, simulations suggest that canopy thinning in wet/warm and dry/cold sites can effectively improve snow water storage and snowpack duration. In nature, because canopy density is correlated with climate such that canopies tend to have a higher density in wet/warm climates than dry/cold climates, canopy thinning is likely more effective in wet/warm climates. Lastly, given a higher snowpack sensitivity to changing canopy density in wet/warm and dry/cold winter climates, locations in these climates can benefit from fine-scale canopy snow modeling.



Water Resources Research



Figure 12. For three sites representative of (a) wet/warm, (b) dry/cold, and (c) wet/cold climate, simulated monthly snowpack energy balance components for water years 2011 and 2012 in the open, high-density forest, medium-density forest, and low-density forest. The period from June to October is excluded when snow is normally thin or not present in either the open or the forest. NSW is net shortwave radiation, NLW is net longwave radiation, H is sensible heat, and LE is latent heat.

| Та | bl | e | 3 |
|----|----|---|---|
| | | | |

Summary of Seasonal Snowpack Energy Balance for Three Representative Sites

| $\sum_{i=1}^{n} j_{i} = $ | | | | | | | | | | |
|---|--------------|----------|----------|----------|----------|----------|----------|----------|----------|----------|
| | | NSW | | | NLW | | | Q | | |
| Canopy condition | Climate zone | Wet cold | Wet warm | Dry cold | Wet cold | Wet warm | Dry cold | Wet cold | Wet warm | Dry cold |
| Open | DJF | 27.2 | 24.0 | 16.6 | -39.8 | -35.7 | -38.0 | 0.1 | 3.0 | 0.1 |
| | MAM | 64.8 | 52.5 | 36.6 | -51.2 | -38.7 | -38.7 | 16.8 | 17.0 | 9.2 |
| Low density | DJF | 17.7 | 16.6 | 10.8 | -20.1 | -19.2 | -13.7 | -2.6 | -3.9 | -0.8 |
| | MAM | 45.5 | 40.2 | 26.0 | -30.2 | -20.5 | -18.6 | 10.5 | 16.3 | 7.3 |
| Medium density | DJF | 6.8 | 5.6 | 4.3 | -6.7 | -0.8 | -3.5 | -5.6 | 2.2 | -0.8 |
| | MAM | 17.7 | 14.1 | 8.9 | -5.7 | -0.7 | -0.3 | 5.1 | 8.5 | 6.4 |
| High density | DJF | 1.6 | 1.3 | 1.2 | 1.1 | 5.8 | 3.4 | -5.9 | 2.8 | -0.3 |
| | MAM | 4.4 | 3.6 | 2.1 | 7.3 | 8.9 | 5.2 | 3.9 | 7.4 | 4.4 |

Note. NSW is net shortwave radiation, NLW is net longwave radiation, and Q is net energy input to snowpack. The unit of all terms is in W/m². DJF represents mean values over December through February, and MAM represents mean values over March through May of water years 2011–2012 of water years 2011–2012.

Data Availability Statement

The DHSVM model and BCQC (bias correction and quality controlled) SNOTEL data can be accessed at https://dhsvm.pnnl.gov/.

Acknowledgments

This research by Pacific Northwest National Laboratory was supported jointly by the U.S. Department of Energy Bioenergy Technologies Office and the Strategic Environmental Research and Development Program. Battelle Memorial Institute operates the Pacific Northwest National Laboratory (PNNL) for the U.S. Department of Energy under contract DE-AC06–76RLO-1830. Lastly, the authors would like to thank Dr. Charlie Luce, Dr. Ernesto Trujillo, Dr. Travis Roth, and two anonymous reviewers for constructive comments improving the manuscript.

References

- Andreadis, K. M., Storck, P., & Lettenmaier, D. P. (2009). Modeling snow accumulation and ablation processes in forested environments. Water Resources Research, 45(5). https://doi.org/10.1029/2008WR007042
- Bales, R. C., Molotch, N. P., Painter, T. H., Dettinger, M. D., Rice, R., & Dozier, J. (2006). Mountain hydrology of the western United States. Water Resources Research, 42(8). https://doi.org/10.1029/2005WR004387
- Barnett, T. P., Adam, J. C., & Lettenmaier, D. P. (2005). Potential impacts of a warming climate on water availability in snow-dominated regions. *Nature*, 438, 303–309. https://doi.org/10.1038/nature04141
- Bohn, T. J., Livneh, B., Oyler, J. W., Running, S. W., Nijssen, B., & Lettenmaier, D. P. (2013). Global evaluation of MTCLIM and related algorithms for forcing of ecological and hydrological models. Agricultural and Forest Meteorology, 176, 38–49. https://doi.org/10.1016/j. agrformet.2013.03.003
- Coughlan, J. C., & Running, S. W. (1997). Regional ecosystem simulation: A general model for simulating snow accumulation and melt in mountainous terrain. *Landscape Ecology*, 12(3), 119–136. https://doi.org/10.1023/A:1007933813251
- Cristea, N. C., Lundquist, J. D., Loheide, S. P., Lowry, C. S., & Moore, C. E. (2014). Modelling how vegetation cover affects climate change impacts on streamflow timing and magnitude in the snowmelt-dominated upper Tuolumne Basin, Sierra Nevada. *Hydrological Processes*, 28(12), 3896–3918. https://doi.org/10.1002/hyp.9909
- Dickerson-Lange, S. E., Gersonde, R. F., Hubbart, J. A., Link, T. E., Nolin, A. W., Perry, G. H., et al. (2017). Snow disappearance timing is dominated by forest effects on snow accumulation in warm winter climates of the Pacific Northwest, United States. *Hydrological Processes*, 31(10), 1846–1862. https://doi.org/10.1002/hyp.11144
- Essery, R., Pomeroy, J., Ellis, C., & Link, T. (2008). Modelling longwave radiation to snow beneath forest canopies using hemispherical photography or linear regression. *Hydrological Processes*, 22(15), 2788–2800. https://doi.org/10.1002/hyp.6930
- Essery, R., Rutter, N., Pomeroy, J., Baxter, R., Stähli, M., Gustafsson, D., et al. (2009). SNOWMIP2: An evaluation of forest snow process simulations. Bulletin of the American Meteorological Society, 90(8), 1120–1136. https://doi.org/10.1175/2009BAMS2629.1
- Friesen, J., Lundquist, J., & Van Stan, J. T. (2015). Evolution of forest precipitation water storage measurement methods. *Hydrological Processes*, 29(11), 2504–2520. https://doi.org/10.1002/hyp.10376
- Gelfan, A. N., Pomeroy, J. W., & Kuchment, L. S. (2004). Modeling forest cover influences on snow accumulation, sublimation, and melt. Journal of Hydrometeorology, 5(5), 785–803. https://doi.org/10.1175/1525-7541(2004)005<0785:MFCIOS>2.0.CO;2
- Hamlet, A. F., & Lettenmaier, D. P. (2007). Effects of 20th century warming and climate variability on flood risk in the western U.S. Water Resources Research, 43(6). https://doi.org/10.1029/2006WR005099
- Harpold, A. A., Molotch, N. P., Musselman, K. N., Bales, R. C., Kirchner, P. B., Litvak, M., & Brooks, P. D. (2015). Soil moisture response to snowmelt timing in mixed-conifer subalpine forests. *Hydrological Processes*, 29(12), 2782–2798. https://doi.org/10.1002/hyp.10400
- Hawthorne, S. N. D., Lane, P. N. J., Bren, L. J., & Sims, N. C. (2013). The long term effects of thinning treatments on vegetation structure and water yield. *Forest Ecology and Management*, 310, 983–993. https://doi.org/10.1016/j.foreco.2013.09.046
- Keith, H., Mackey, B. G., & Lindenmayer, D. B. (2009). Re-evaluation of forest biomass carbon stocks and lessons from the world's most carbon-dense forests. *Proceedings of the National Academy of Sciences*, 106(28), 11635–11640. https://doi.org/10.1073/pnas.0901970106
- Kobayashi, D. (1987). Snow accumulation on a narrow board. Cold Regions Science and Technology, 13(3), 239–245. https://doi.org/10.1016/0165-232X(87)90005-X
- Lawler, R. R., & Link, T. E. (2011). Quantification of incoming all-wave radiation in discontinuous forest canopies with application to snowmelt prediction. *Hydrological Processes*, 25(21), 3322–3331. https://doi.org/10.1002/hyp.8150
- Liu, Y., Yu, G., Wang, Q., & Zhang, Y. (2014). How temperature, precipitation and stand age control the biomass carbon density of global mature forests. *Global Ecology and Biogeography*, 23(3), 323–333. https://doi.org/10.1111/geb.12113



- Livneh, B., Rosenberg, E. A., Lin, C., Nijssen, B., Mishra, V., Andreadis, K. M., et al. (2013). A long-term hydrologically based dataset of land surface fluxes and states for the conterminous United States: Update and extensions. *Journal of Climate*, 26(23), 9384–9392. https://doi. org/10.1175/JCLI-D-12-00508.1
- Loader, C. (2020). locfit: Local regression, likelihood and density estimation. R package version 1.5-9.4. Retrieved from https://CRAN.R-project.org/package=locfit
- Luce, C. H., Lopez-Burgos, V., & Holden, Z. (2014). Sensitivity of snowpack storage to precipitation and temperature using spatial and temporal analog models. *Water Resources Research*, 50(12), 9447–9462. https://doi.org/10.1002/2013WR014844
- Lundquist, J. D., Dickerson-Lange, S. E., Lutz, J. A., & Cristea, N. C. (2013). Lower forest density enhances snow retention in regions with warmer winters: A global framework developed from plot-scale observations and modeling. *Water Resources Research*, 49(10), 6356–6370. https://doi.org/10.1002/wrcr.20504
- Malle, J., Rutter, N., Mazzotti, G., & Jonas, T. (2019). Shading by trees and fractional snow cover control the subcanopy radiation budget. *Journal of Geophysical Research: Atmospheres*, 124(6), 3195–3207. https://doi.org/10.1029/2018JD029908
- Marks, D., & Dozier, J. (1992). Climate and energy exchange at the snow surface in the alpine region of the Sierra Nevada: 2. Snow cover energy balance. Water Resources Research, 28(11), 3043–3054. https://doi.org/10.1029/92WR01483
- Martin, K. A., Van Stan, J. T., Dickerson-Lange, S. E., Lutz, J. A., Berman, J. W., Gersonde, R., & Lundquist, J. D. (2013). Development and testing of a snow interceptometer to quantify canopy water storage and interception processes in the rain/snow transition zone of the North Cascades, Washington, USA. Water Resources Research, 49(6), 3243–3256. https://doi.org/10.1002/wrcr.20271
- Maurer, E. P., Wood, A. W., Adam, J. C., Lettenmaier, D. P., & Nijssen, B. (2002). A long-term hydrologically based dataset of land surface fluxes and states for the conterminous United States. *Journal of Climate*, 15(22), 3237–3251. https://doi.org/10.1175/1520-0442(2002)015< 3237:althbd>2.0.co;2
- Mazzotti, G., Currier, W. R., Deems, J. S., Pflug, J. M., Lundquist, J. D., & Jonas, T. (2019). Revisiting snow cover variability and canopy structure within forest stands: Insights from airborne lidar data. Water Resources Research, 55(7), 6198–6216. https://doi.org/10.1029/2019WR024898
- Mazzotti, G., Essery, R., Moeser, C. D., & Jonas, T. (2020). Resolving small-scale forest snow patterns using an energy balance snow model with a one-layer canopy. *Water Resources Research*, 56(1). https://doi.org/10.1029/2019WR026129
- Mazzotti, G., Malle, J., Barr, S., & Jonas, T. (2019). Spatially continuous characterization of forest canopy structure and subcanopy irradiance derived from handheld radiometer surveys. *Journal of Hydrometeorology*, 20(7), 1417–1433. https://doi.org/10.1175/JHM-D-18-0158.1
- Mazzotti, G., Webster, C., Essery, R., & Jonas, T. (2021). Increasing the physical representation of forest-snow processes in coarse-resolution models: Lessons learned from upscaling hyper-resolution simulations. *Water Resources Research*, 57(5). https://doi.org/10.1029/2020WR029064
- Mote, P. W., Hamlet, A. F., Clark, M. P., & Lettenmaier, D. P. (2005). Declining mountain snowpack in western North America. Bulletin of the American Meteorological Society, 86(1), 39–49. https://doi.org/10.1175/BAMS-86-1-39
- Murray, C. D., & Buttle, J. M. (2003). Impacts of clearcut harvesting on snow accumulation and melt in a northern hardwood forest. *Journal of Hydrology*, 271(1–4), 197–212. https://doi.org/10.1016/S0022-1694(02)000352-9
- Musselman, K. N., Molotch, N. P., & Brooks, P. D. (2008). Effects of vegetation on snow accumulation and ablation in a mid-latitude sub-alpine forest. *Hydrological Processes*, 22(15), 2767–2776. https://doi.org/10.1002/hyp.7050
- Nash, J. E., & Sutcliffe, J. V. (1970). River flow forecasting through conceptual models part I: A discussion of principles. *Journal of Hydrology*, 10(3), 282–290. https://doi.org/10.1016/0022-1694(70)90255-6
- North, M., Chen, J., Oakley, B., Song, B., Rudnicki, M., Gray, A., & Innes, J. (2004). Forest stand structure and pattern of old-growth western hemlock/Douglas-fir and mixed-conifer forests. *Forest Science*, 50(35), 299–311. https://doi.org/10.1093/forestscience/50.3.299
- Peixoto, J. P., & Oort, A. H. (1992). *Physics of climate* (1st ed.). American Institute of Physics.
- Pomeroy, J. W., Marks, D., Link, T., Ellis, C., Hardy, J., Rowlands, A., & Granger, R. (2009). The impact of coniferous forest temperature on incoming longwave radiation to melting snow. *Hydrological Processes*, 23(17), 2513–2525. https://doi.org/10.1002/hyp.7325
- Prata, A. J. (1996). A new long-wave formula for estimating downward clear-sky radiation at the surface. Quarterly Journal of the Royal Meteorological Society, 122(533), 1127–1151. https://doi.org/10.1002/qj.49712253306
- Roth, T., & Nolin, A. (2017). Forest impacts on snow accumulation and ablation across an elevation gradient in a temperate montane environment. Hydrology and Earth System Sciences, 21(11), 5427–5442. https://doi.org/10.5194/hess-21-5427-2017
- Roth, T. R., & Nolin, A. W. (2019). Characterizing maritime snow canopy interception in forested mountains. Water Resources Research, 55, 2018WR024089. https://doi.org/10.1029/2018WR024089
- Rutter, N., Essery, R., Pomeroy, J., Altimir, N., Andreadis, K., Baker, I., et al. (2009). Evaluation of forest snow processes models (SnowMIP2). *Journal of Geophysical Research*, *114*(D6), D06111. https://doi.org/10.1029/2008JD011063
- Serreze, M. C., Clark, M. P., Armstrong, R. L., McGinnis, D. a, & Pulwarty, R. S. (1999). Characteristics of the western United States snowpack from snowpack telemetry(SNOTEL) data. *Water Resources Research*, 35(7), 2145–2160. https://doi.org/10.1029/1999WR900090
- Sharma, A., Wasko, C., & Lettenmaier, D. P. (2018). If precipitation extremes are increasing, why aren't floods? Water Resources Research, 54(11), 8545–8551. https://doi.org/10.1029/2018WR023749
- Stegen, J. C., Swenson, N. G., Enquist, B. J., White, E. P., Phillips, O. L., Jørgensen, P. M., et al. (2011). Variation in above-ground forest biomass across broad climatic gradients. *Global Ecology and Biogeography*, 20(5), 744–754. https://doi.org/10.1111/j.1466-8238.2010.00645.x
- Storck, P. (2000). Trees, snow, and flooding: An investigation of forest can- opy effects on snow accumulation and melt at the plot and watershed scales in the Pacific Northwest. Water resources series technical report No. 161. Retrieved from https://ir.library.oregonstate.edu/concern/ defaults/xk81jr150
- Storck, P., Bowling, L., Wetherbee, P., & Lettenmaier, D. (1998). Application of a GIS-based distributed hydrology model for prediction of forest harvest effects on peak stream flow in the Pacific Northwest. *Hydrological Processes*, 12(6), 889–904. https://doi.org/10.1002/ (sici)1099-1085(199805)12:6<889::aid-hyp661>3.0.co;2-p
- Storck, P., Lettenmaier, D. P., & Bolton, S. M. (2002). Measurement of snow interception and canopy effects on snow accumulation and melt in a mountainous maritime climate, Oregon, United States. *Water Resources Research*, 38(11), 55–116. https://doi.org/10.1029/2002WR001281 Sturm, M., Goldstein, M. A., & Parr, C. (2017). Water and life from snow: A trillion dollar science question. *Water Resources Research*, 53(5),
- 3534–3544. https://doi.org/10.1002/2017WR020840 Sun, N., Yan, H., Wigmosta, M. S., Leung, L. R., Skaggs, R., & Hou, Z. (2019). Regional snow parameters estimation for large-domain hydrological applications in the western United States. *Journal of Geophysical Research: Atmospheres*, 124(10), 5296–5313. https://doi. org/10.1029/2018JD030140
- Sun, N., Yearsley, J., Voisin, N., & Lettenmaier, D. P. (2015). A spatially distributed model for the assessment of land use impacts on stream temperature in small urban watersheds. *Hydrological Processes*, 29(10), 2331–2345. https://doi.org/10.1002/hyp.10363
- Thomas, S. C., & Winner, W. E. (2000). Leaf area index of an old-growth Douglas-fir forest estimated from direct structural measurements in the canopy. Canadian Journal of Forest Research, 30(12), 1922–1930. https://doi.org/10.1139/x00-121

- Thornton, P. E., & Running, S. W. (1999). An improved algorithm for estimating incident daily solar radiation from measurements of temperature, humidity, and precipitation. *Agricultural and Forest Meteorology*, 93(4), 211–228. https://doi.org/10.1016/S0168-1923(98)00126-9
- Thyer, M., Beckers, J., Spittlehouse, D., Alila, Y., & Winkler, R. (2004). Diagnosing a distributed hydrologic model for two high-elevation forested catchments based on detailed stand- and basin-scale data. *Water Resources Research*, 40(1). https://doi.org/10.1029/2003WR002414
- Veatch, W., Brooks, P. D., Gustafson, J. R., & Molotch, N. P. (2009). Quantifying the effects of forest canopy cover on net snow accumulation at a continental, mid-latitude site. *Ecohydrology*, 2(2), 115–128. https://doi.org/10.1002/eco.45
 Wostrick, K., Storek, P., & Mass, C. (2002). Description and avaluation of a hydrometagraphytical foresect system for mountainous watershede.
- Westrick, K., Storck, P., & Mass, C. (2002). Description and evaluation of a hydrometeorological forecast system for mountainous watersheds. Weather and Forecasting, 17, 250–262. https://doi.org/10.1175/1520-0434(2002)017<0250:daeoah>2.0.co;2
- Wheeler, J. A., Cortés, A. J., Sedlacek, J., Karrenberg, S., van Kleunen, M., Wipf, S., et al. (2016). The snow and the willows: Earlier spring snowmelt reduces performance in the low-lying alpine shrub Salix herbacea. *Journal of Ecology*, *104*(4), 1041–1050. https://doi. org/10.1111/1365-2745.12579
- Wigmosta, M. S., Nijssen, B., Storck, P., & Lettenmaier, D. (2002). The distributed hydrology soil vegetation model. In V. P. Singh (Ed.), Mathematical models of small watershed hydrology and applications (pp. 7–42). Water Resources Publication.
- Wigmosta, M. S., Vail, L. W., & Lettenmaier, D. P. (1994). A distributed hydrology-vegetation model for complex terrain. Water Resources Research, 30(6), 1665–1679. https://doi.org/10.1029/94wr00436
- Winkler, R. D., Spittlehouse, D. L., & Golding, D. L. (2005). Measured differences in snow accumulation and melt among clearcut, juvenile, and mature forests in southern British Columbia. *Hydrological Processes*, 19(1), 51–62. https://doi.org/10.1002/hyp.5757
- Yan, H., Sun, N., Wigmosta, M., Skaggs, R., Hou, Z., & Leung, L. R. (2018). Next-generation intensity-duration-frequency curves for hydrologic design in snow-dominated environments. *Water Resources Research*, 54(2), 1093–1108. https://doi.org/10.1002/2017WR021290
- Yan, H., Sun, N., Wigmosta, M., Skaggs, R., Hou, Z., & Leung, L. R. (2019). Next-generation intensity-duration-frequency curves to reduce errors in peak flood design. *Journal of Hydrologic Engineering*, 24(7), 04019020. https://doi.org/10.1061/(ASCE)HE.1943-5584.0001799
- Yan, H., Sun, N., Wigmosta, M., Skaggs, R., Leung, L. R., Coleman, A., & Hou, Z. (2019). Observed spatiotemporal changes in the mechanisms of extreme water available for runoff in the western United States. *Geophysical Research Letters*, 46(2), 767–775. https://doi.org/10.1029/2018GL080260